Author’s response to reviews

Title: How do older patients and their GPs evaluate shared decision-making in health care?

Authors:

   Danica Rotar-Pavlic (danica.rotar@guest.arnes.si)
   Igor Svab (igor.svab@mf.uni-lj.si)
   Raymond Wetzels (r.wetzels@kwazo.umcn.nl)

Version: 3 Date: 26 February 2008

Author’s response to reviews: see over
Dear editor,

The referee has requested some further minor revisions to our manuscript. On behalf of the authors I am sending the revision of the article *How do older patients and their GPs evaluate shared decision-making in healthcare?* The discussion section is rewritten. The nursing literature and therapy research is included and cited. The diversity of the area of patient participation is underlined (some patients want the GP to make decisions for them). There are evidences of the importance of developing a relationship before carrying out interventions.

Yours sincerely,

Danica Rotar Pavlič